Author's response to reviews

Title: Health worker preferences for working in rural clinics in Uganda: a discrete choice experiment

Authors:

Peter C Rockers (prockers@hsph.harvard.edu)
Wanda Jaskiewicz (wjaskiewicz@capacityplus.org)
Laura Wurts (laurawurts@gmail.com)
Margaret E Kruk (mkruck@columbia.edu)
George Mgomella (mgomella@gmail.com)
Francis Ntalazi (francisntalazi@yahoo.com)
Kate Tulenko (ktulenko@capacityplus.org)

Version: 2 Date: 12 December 2011

Author's response to reviews:

Dear Editors:

Please accept this resubmitted draft of our manuscript. We have made one revision, as requested. We now include the survey instruments as 'Additional Files' and reference these instruments in the Methods section of the manuscript.

Sincerely,

Peter C. Rockers